Author's response to reviews

Title: Ecological study of socio-economic indicators and prevalence of asthma in schoolchildren in urban Brazil.

Authors:

Sergio S da Cunha (cunhass@ufba.br)
Mar R Pujades (mar.pujades@epicentre.msf.org)
Mauricio L Barreto (mauricio@ufba.br)
Bernd Genser (bernd.genser@bgstats.com)
Laura C Rodrigues (laura.rodrigues@lshtm.ac.uk)

Version: 6 Date: 20 July 2007

Author's response to reviews:

Formatting changes completed
Affiliation written in full, and "UK" substituted for "England" as requested.